Author's response to reviews

Title: The null hypothesis significance test in health sciences research (1995-2006): statistical analysis and interpretation

Authors:

Luis Carlos Silva-Ayçaguer (lcsilva@infomed.sld.cu)
Patricio Suárez-Gil (patricio.suarez@sespa.princast.es)
Ana Fernández-Somoano (afsomo@yahoo.es)

Version: 4 Date: 3 May 2010

Author's response to reviews: see over
Dear Colleagues of The BioMed Central Editorial Team:

We have received your note dated 2010/4/21 concerning our paper The Null Hypothesis Significance Test in Health Sciences Research (1995-2006): Statistical Analysis and Interpretation (MS: 2878749073306160) and your request of revision of the manuscript to ensure that the language is at an adequate standard for publication as well as its conformity to the journal style.

Once revised both aspects, language by two English native speakers (Tom Piazza, University of California at Berkeley, CA, USA; and Ann Séror, Laval University, Quebec City, Canada) and style according to the journal, we submit the new version of our paper.

Yours sincerely,

Patricio Suárez-Gil